HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY COMP 102 : Pr ogramming & Computer Fundamentals I Summer 1996 Laboratory Assignment 2 8 & 10 July

For each of the problems given below, 1) Design an algorithm for solving the problem. 2) Implement your solution using C++.

PROBLEM #1 Write a program which reads in two positive integers upperbound and factor (in this order) and prints all the integers between 1 and upperbound (inclusive) which are divisible by factor. Example:

Please enter the upperbound: 20 Please enter the factor: 5 The numbers are: 5 10 15 20

PROBLEM #2 Write a program which reads in a positive integer upperbound and prints all the prime numbers between 1 and upperbound (inclusive). You can compute these prime numbers by doing the following. Iterate a variable N from 1 to upperbound. N is a prime number if it is NOT divisible by any number between 2 and N-1 (inclusive). Recall that 1 is NOT a prime number. Example:

Please enter the upperbound: 30 The prime numbers are: 2 3 5 7 11 13 17 19 23 29